**Author's response to reviews**

**Title:** Facility and home based HIV Counseling and Testing: a comparative analysis of uptake of services by rural communities in southwestern Uganda

**Authors:**

Edgar M Mulogo (emulogo2000@gmail.com)  
Abdulaziz S Aden (aziz@iss.it)  
Ranieri Guerra (ranieri.guerra@iss.it)  
Sebastian O Baine (sbaine@musph.ac.ug)

**Version:** 4  **Date:** 21 February 2011

**Author's response to reviews:** see over
17 February 2011

The Editor
BMC Health Services Research Journal

Dear Sir/Madam,

Re: Submission of Edited Manuscript (MS: 3546848974024230)

Please find attached a manuscript entitled “Facility and home based HIV Counseling and Testing: a comparative analysis of uptake of services by rural communities in southwestern Uganda”. I have edited the manuscript as requested by the BMC Editorial team. The edits made are given here.

1. I have edited the names of the author on the manuscript to the preferred style (‘First name Initial Last name).
2. I have removed all the highlighting from the manuscript.
3. Typography:
   a. I have standardized the appearance all the “brackets” in the manuscript (there were some sections with square brackets)
   b. I have also standardized the use of commas to separate the lower and upper limits of the 95% confidence intervals (CI)

Hoping for your kind consideration

Yours sincerely,

Dr. Edgar Mugema Mulogo
Corresponding author/Senior Lecturer
Dept. of Community Health
MUST